Author's response to reviews

Title: Membranous Nephropathy and Lupus like syndrome after Haematopoietic Cell Transplantation: A Case Report and Literature Review

Authors:

Kostas Stylianou (kstylianu@gmail.com)
Stavros Stratakis (indian_steven@yahoo.gr)
Vasiliki Mavroeidi (y_mavroidi@yahoo.gr)
Ioannis Petrakis (petrakgia@gmail.com)
Dimitris Xydakis (dxydakis@gmail.com)
Eleftheria Vardaki (elevard@in.gr)
Spyros Stratigis (stratigiss@yahoo.com)
Kostas Perakis (perkost@gmail.com)
Theodora Katsarou (theokats@windowslive.com)
Peggy Kanellou (kanelloup@yahoo.gr)
Irene Xylouri (antonakisnikos@yahoo.gr)
Constantina Petraki (nelniko@otenet.gr)
Michael Alexandrakis (alexandm@med.uoc.gr)
Eugene Daphnis (eugenedaphnis@yahoo.gr)

Version: 3 Date: 29 April 2010

Author's response to reviews: see over
On behalf of my co-authors I would like to thank you for your reply concerning our manuscript, entitled “Membranous Nephropathy and Lupus like syndrome after Haematopoietic Cell Transplantation: A Case Report and Literature Review”.

We have carefully considered the concerns raised by the editorial office and have responded to each comment accordingly. Changes in the revised manuscript are indicated in red characters.

Point-by-point answer to the editorial office comments.

"There are still some minor changes that need to be changed:

MN in introduction needs to be defined.
Abbreviations like PR, CR need to be removed as these are not standard medical abbreviations and are confusing
TTP - needs to be defined

Response: Indicated changes have been made in the revised manuscript

“Lastly the English spelling of anasarca is correct but fully agree with our Greek compatriots that the correct spelling from the origins is also correct, but the English have a habit of borrowing words other languages and changing the original spelling!"

Response: Certainly anasarca is the right spelling as we also indicate in our previous response to the reviewer. Actually there was a misunderstanding. The reviewer’s comment was that “anasarca” has to be written with a “c”, which made us think that he meant
“anacarca” which is obviously incorrect. On the other hand, I was unpleasantly surprised when I realized that despite all that discussion I kept misspelling the word in the main manuscript. The correction has been made now.

“In addition to the concerns raised by the Associate Editor, please also address the following editorial requests:

Please restructure the abstract into the following three sections:
Introduction, Case Presentation and Conclusion”

Response: The indicated changes have been made in the revised manuscript

We trust that this revised version now meets with your approval.

Assuring you of our best attention at all times,

Dr Kostas Stylianou

Consultant Nephrologist